Author's response to reviews

Title: Quantitative assessment of Vigabatrin-attributable visual field loss using semi-automated kinetic perimetry

Authors:

Katarzyna Nowomiejska (katarzynanowomiejska@mailcity.com)
Konrad Rejdak (konrad.rejdak@umlub.pl)
Marian Jedrych (marian.jedrych@umlub.pl)
Tomasz Zarnowski (zarnowski@poczta.onet.pl)
Katarzyna Ksiazek (kasiakal75@wp.pl)
Piotr Ksiazek (zdrowie.publiczne@umlub.pl)
Ryszard Maciejewski (ryszard.maciejewski@umlub.pl)
Anselm Juenemann (Anselm.Juenemann@uk-erlangen.de)
Ulrich Schiefer (ulrich.schiefer@htw-aalen.de)
Robert Rejdak (robertrejdak@yahoo.com)

Version: 3  Date: 20 November 2013

Author's response to reviews: see over
Dear Editor of „BMC Neurology”,

Please find below changes made to the manuscript “Quantitative assessment of Vigabatrin-attributable visual field loss using semi-automated kinetic perimetry” by Katarzyna Nowomiejska, Konrad Rejdak, Marian Jedrych, Tomasz Zarnowski, Katarzyna Książek, Piotr Książek, Ryszard Maciejewski, Anselm Juenemann, Ulrich Schiefer and Robert Rejdak.

1. In the abstract “Purpose” was replaced by “Background”.
2. In the manuscript text “Introduction” was replaced by “Background”.
3. In Methods chapter it as added: Ethics Committee of the Medical University in Lublin.
4. There is one figure 7 containing 5 figures.
5. The table is included after References section, there is no text in vertical direction inside the table.

We hope that now the manuscript is ready for review process,

Kind regards,

Katarzyna Nowomiejska